Author's response to reviews

Title: Safety and tolerability of donepezil 23 mg in moderate to severe Alzheimer's disease

Authors:

Martin Farlow (mfarlow@iupui.edu)
Felix Veloso (drfveloso@sasktel.net)
Margaret Moline (margaret_moline@eisai.com)
Jane Yardley (Jane_Yardley@eisai.net)
Elimor Brand-Schieber (Elimor_Brand-Schieber@eisai.com)
Francesco Bibbiani (Francesco_Bibbiani@eisai.com)
Heng Zou (Heng_Zou@Eisai.com)
Timothy Hsu (Timothy_Hsu@eisai.com)
Andrew Satlin (Andrew_Satlin@eisai.com)

Version: 3 Date: 29 April 2011

Author's response to reviews:

Dear Editor,

All revisions requested by you in your e-mail dated April 17, 2011 have been made.

Figure 2 was deleted as it was determined to be redundant to data presented in Table 6.

No other changes were made to the manuscript.

Thank you,

Martin Farlow, MD